Course Information Mathematics 477, Section 03

Mathematical Theory of Probability Spring Semester 2003

Weekly Schedule Day Period Location Monday 6 HICK 103 Wednsday 6 HICK 103

Instructor: Yi-Zhi Huang Office: Hill 332 (Busch) Telephone: 445-1314 Office Hours: M 2:30–3:30 pm, W 2:30–3:30 pm or by appointment

Examination Dates: First Hour Examination Second Hour Examination Final Examination Both of the hour examinations are held at the regular class time and in the regular meeeting room. The location of the final examination will be announced several weeks before the end of the term. Homework problems with * are due in next Monday’s classes. No late homework will be accepted.

Text: A First Course in Probability by Sheldon Ross, 6^{th} edition, Prentice Hall (2002).

Grading Policy: the various components of the course are weighted as follows in the determination of your course grade.

Exam 1 20% Exam 2: 20% Quizzes and homework: 20% Final exam: 40%